Author’s response to reviews

Title: Plasma long non-coding RNA BACE1 as a novel biomarker for diagnosis of Alzheimer disease

Authors:
Liang Feng (fengliangyw@126.com)
Yu-Ting Liao (1072619714@qq.com)
Jin-Cai He (hjc@wmu.edu.cn)
Cheng-Long Xie (xiechenglong1987@126.com)
Si-Yan Chen (614803555@qq.com)
Hui-Hui Fan (hmwu@wmu.edu.cn)
Zhi-Peng Su (drszhipeng@163.com)
Zhen Wang (13736716097@139.com)

Version: 3 Date: 25 Oct 2017

Author’s response to reviews:

Dear editor

We are writing to re-submit the revised manuscript entitled “Plasma long non-coding RNA BACE1 as a novel biomarker for diagnosis of Alzheimer disease” (NURL-D-17-00245). I would like to take this opportunity to thank the reviewers to provide invaluable comments for the improvement of our manuscript. We have carefully studied the comments of the reviewers and made necessary changes in the revised manuscript. We hope that we have addressed all of the questions satisfactorily and the manuscript at current version can be accepted in your journal. I am looking forward to receiving your favor reply. Thank you very much.

Yours faithfully
Reviewer reports:

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing service.

RE: We are so sorry for our incorrect writing. We have tried our best effort to check the entire manuscript again, especially in phrasing trouble, type mistakes, grammatical errors, awkward sentences and organisation of the discussion in the revised manuscript. These changes will not influence the content and framework of the paper. Meanwhile, we gratefully acknowledge Dr. Man-Man Zhang from Wenzhou Medical University for the language improvement of the paper.